Author's response to reviews

Title: Peripheral T-lymphocytes express WNT7A and its restoration in leukemia-derived lymphoblasts inhibits cell proliferation

Authors:

Alejandra B Ochoa-Hernández (bereniceqfb@hotmail.com)
Ivan D Meza-Canales (idavidmc@hotmail.com)
Mónica A Rosales-Reynoso (mareynos@hotmail.com)
Beatriz García-Castro (gacas.bet@hotmail.com)
Moisés Ramos-Solano (leboyfr@hotmail.com)
Judith A Rosales-Aviña (jjudithabisaq1@gmail.com)
Esperanza Barrera-Chairez (vivibarrera2010@hotmail.com)
Alejandro Bravo-Cuellar (abrarvocster@gmail.com)
Luis F Jave-Suarez (lfjave@yahoo.com)
Patricio Barros-Núñez (pbarros_gdl@yahoo.com.mx)
Adriana Aguilar-Lemarroy (adry.aguilar.lemarroy@gmail.com)
Pablo C Ortíz-Lazareno (paulcesar05@hotmail.com)
Georgina Hernández-Flores (geodic1967@yahoo.com.mx)

Version: 2 Date: 8 December 2011

Author's response to reviews: see over
Dear Editorial Board Members,

BMC Cancer

We have now revised our manuscript (*Peripheral T-lymphocytes express WNT7A and its restoration in leukemia-derived lymphoblasts inhibits cell proliferation*) according to the comments and suggestions raised by the reviewers, which we have found very helpful for the improvement of the manuscript.

We have attempted to address all of the concerns addressed by the reviewers carefully. We added new information that supports the affirmations we have made.

Additionally, we attempted to clarify the presentation of the manuscript. We have made corrections in the writing throughout the manuscript. Furthermore, the manuscript was proofread by a native English speaker.

We hope that we have responded to all of the questions generated by the reviewers and we are grateful for all of the effort that the reviewers generously invested in this paper.

Thanks in advance,

Adriana Aguilar Lemarroy, PhD
(Corresponding Autor)
Sierra Mojada No. 800,
Col. Independencia,
C.P. 44340, Guadalajara, Jalisco,
México
Tel: +52 33 36 17 00 60 ext. 31926
Fax: +52 33 36 54 08 54
email: adry.aguilar.lemarroy@gmail.com